Author's response to reviews

Title: A Computer-Based Medical Record System and Personal Digital Assistants to Assess and Follow Patients with Respiratory Tract Infections Visiting a Rural Kenyan Health Centre

Authors:

Lameck Diero (ldiero@africaonline.co.ke)
Joseph K Rotich (jkarotich@yahoo.co.uk)
John Bii (jbii@mtrh.org)
Burke W Mamlin (bmamlin@iupui.edu)
Robert M Einterz (reinterz@iupui.edu)
Irene Z. Kalamai (izkalamai@yahoo.com)
William M Tierney (wtierney@iupui.edu)

Version: 4 Date: 3 April 2006

Author's response to reviews:

The following formatting changes have been made:
Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.
Done.
Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.
Done.
E-mail address - Please will you replace jkarotich@yahoo.com with an active e-mail address.
Dr. Rotich's real (and only) e-mail address is jkaroich@yahoo.co.uk.
Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).
Done.
Author list - Please remove titles from the list of authors.
Done.
Remove text - Please remove "2854 words of text, 20 references, 2 tables, and 5 figures" and "[Abstract = 216 words]" from the text.
Done.
Underlining - Please do not underline any section headings.
Done.
Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals).
Done.
Introduction - Please rename this section 'Background'.
Done.
References - The reference list should contain all authors' names, regardless of how many there are. The term "et al" should not be used.
Done.
Figure legends - please change
Figure 3A: PDA in use.
Figure 3B: Screen used to collect data on cough.
Figure 3C: Screen used to collect data on sputum production.
To:
Figure 3. The programmed PDA and sample screens. A: PDA in use. B: Screen used to collect data on cough. C: Screen used to collect data on sputum production.
Done.